Author’s response to reviews

Title: The Validation of Organisational Culture Assessment Instrument in Healthcare Setting: Results from a Cross-sectional Study in Vietnam

Authors:

Huy Nguyen (nvanhuy@yahoo.com)
Thu Nguyen (nguyenhoaithu@hmu.edu.vn)
Anh Nguyen (dennguyenle@gmail.com)
Au Nguyen (haiau04sh@gmail.com)
Phuong Nguyen (nguyenthephuong.hmu@gmail.com)
Cham Nguyen (Cham.nguyenthi26@gmail.com)
Minh Pham (minh.pham@burnet.edu.au)

Version: 6 Date: 06 Feb 2020

Author’s response to reviews:

To Editors,
BMC Women’s Health

Re: “Perceived Barriers to Early Diagnosis of Breast Cancer in South and Southwestern Ethiopia: a qualitative study”

Thank you very much for your email dated February 2nd, 2020 with additional final review/editorial comments provided. We have accordingly made the requested edits of the manuscript and made necessary revisions on the paper. For your kind consideration, please find the following documents submitted:

1) A new revised and clean version of the manuscript, and

2) A “point by point” response to the comments and suggestions of the assistant editor.

We hope that you will find the edits as per your expectation and we look forwards to hear from you soon.

Yours Sincerely,
Dr. Adamu Addissie
Email: adamuaddissie@gmail.com
Point by Point Responses

1. Thank you for amending the ages to age ranges in the quotes from the participants. Please also change the ages to age ranges in the quotes from the healthcare providers. Please ensure all quotes do not contain specific ages.

Response: Thanks for the feedback. All ages of health care providers are not put in ranges similar to the patients.

2. Please note, the "Availability of Data and Materials" declaration refers to the raw data used in your study. The statement currently provided is not correct in the case of your manuscript. We strongly encourage all authors to share their raw data, either by providing it in a supplementary file or depositing it in a public repository and providing the details on how to access it in this section. If you do not wish to share your data, please clearly state this in this section along with a justification.

Response: Thank you for the feedback and this is now corrected as follows. The raw data used and analysed for the current study is deposited by the authors in the repository of the Non Communicable Diseases (NCD) Research Working Group of Addis Ababa University. The de-identified raw data is available up on reasonable request. Please email adamu.addissie@aau.edu.et for any data requests.

3. If abbreviations are used in the text they should be defined in the text at first use, and a list of abbreviations should be provided before the Declarations.

Response: We have checked and verified that all abbreviations are now defined in first use.

4. At this stage, please upload your manuscript as a single, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files.

Response: all versions now uploaded as clean versions